# COUNTING THE COST PART 2 ECONOMIC COSTS

The current and future burden of arthritis

Report prepared for Arthritis Australia May 2016



**The University of Sydney** Deborah Schofield Rupendra Shrestha Michelle Cunich

### Foreword

#### About this report:

Counting the Cost: The Current and Future Burden of Arthritis provides up to date estimates of the prevalence and costs of arthritis from 2015 to 2030 under a 'business as usual' scenario. It also models the impact of implementing non-surgical interventions for knee osteoarthritis to illustrate the potential benefits for individuals, governments and the economy from better arthritis care.

Part 1 Healthcare Costs estimates the current and future prevalence of arthritis and the associated costs to the health system. Part 2 Economic Costs considers the economic costs of arthritis, including lost personal income, increased welfare payments, reduced taxation revenue and lost GDP.



Arthritis is one of the most common, costly and disabling chronic conditions in Australia. In its many forms, it affects nearly four million people of all ages, including children.

The personal, social and economic costs of arthritis, as highlighted in this report, are immense but tend to be poorly

recognised. These costs amount to many billions of dollars a year and include health care costs, lost personal income and national productivity losses from reduced work capacity due to arthritis and, of course, the immeasurable cost of lost wellbeing.

With arthritis prevalence set to reach 5.4 million people by 2030, associated costs will continue to grow, putting increasing pressure on the sustainability of the health, welfare and aged care systems.

Yet much can be done to alleviate these costs by implementing simple programs to prevent and better manage arthritis. As highlighted in this report, a simple intervention for knee osteoarthritis could achieve savings to the health system of over \$170 million a year, as well as helping to keep more people in the work force, yielding additional economic benefits.

The time has come to give programs that provide better care and support for people with arthritis the priority they deserve. With the health and welfare system costs of arthritis set to grow by more than \$150 million a year, we simply can't afford not to.

Ainslie Cahill CEO Arthritis Australia

# Table of contents

Fo	rewor	ď		ii
Tal	ole of	content	5	iii
Lis	t of ta	ables		iv
Ac	know	ledgeme	ents	V
Ex	ecutiv	e summa	ary	vi
1.	Back	ground		1
	1.1	Objecti	ves	2
	1.2	Overvie	w of model development	3
	1.3	Metho	dology	3
	1.4	Structu	re of the report	4
2.		econom 1 2015 t	ic costs of arthritis through lost productivity on individuals, government and society o 2030	5
	2.1	Lost lab	pour force participation due to arthritis	5
	2.2	The eco	pnomic costs	5
	2.3	Measu	rement of economic costs	5
	2.4	Statistic	cal simulation	6
	2.5	Results		6
3.	The	costs of	arthritis through lost productivity due to informal caregiving from 2015 to 2030	12
	3.1	Metho	ds	13
	3.2	Results		13
4.	Disc	ussion a	nd concluding remarks	15
	4.1	Costs c	f arthritis through lost labour force participation	15
	4.2	Costs c	f arthritis through informal caregiving	16
5.			e gains and the associated economic gains if people with osteoarthritis of the knee gaged in the ADAPT trial strategies	18
	5.1	Data ar	nd methods	19
		5.1.1	Effectiveness of interventions	19
		5.1.2	Association between SF-36 bodily pain scores and labour force participation	19
		5.1.3	Labour force impacts	20
		5.1.4	Economic benefits	20
	5.2	Results		20
		5.2.1	Not in the labour force due to knee OA	20
		5.2.2	Association between SF-36 bodily pain scores and labour force participation	21
		5.2.3	Impacts on labour force participation	22
		5.2.4	Economic impacts	22
6.	Con	clusions		24
7.	Refe	erences		25

## List of tables

Table 1.	Mean and median weekly income, welfare payments, taxes and total value of savings and assets of individuals with and without arthritis, Australian population aged 15-64 years (in 2015 Australian dollars)	7-8
Table 2.	Mean differences in weekly income, welfare payments, taxes and total value of savings and assets between people with lost productivity due to arthritis and those employed full-time without arthritis and those employed part-time without arthritis (adjusted for age, sex and highest level of education), Australian population aged 15-64 years (in 2015 Australian dollars)	10
Table 3.	National economic costs of lost labour force participation due to arthritis per year, Australian population aged 15-64 years (in 2015 Australian dollars, in millions)	11
Table 4.	Mean and median weekly income, welfare payments, taxes of primary carers of people with arthritis as main disabling condition, Australian population aged 15-64 years (in 2015 AU\$)	14
Table 5.	Projected numbers of those not in the labour force due to knee OA, aged 15-64 years	21
Table 6.	Percentage increase in the labour force participation rates associated with pain reduction due to Diet plus Exercise intervention for the management of knee OA in the ADAPT study	21
Table 7.	Estimated number of people who would have avoided being out of the labour force due to their knee OA as a result of dietary weight loss plus exercise intervention for the management of their knee OA, 15-64 years old	22
Table 8.	Estimated economic benefits from increased labour force associated with reduced pain as a result of managing their knee OA with dietary weight loss plus exercise intervention, 15-64 years old (in thousands, in 2015 AU\$)	23

### Acknowledgements

This research was supported by funding from Arthritis Australia. This funding was made possible by an unrestricted grant from The Medicines Australia Community Chest to Arthritis Australia. The Community Chest supporters are AbbVie Pty Ltd, AstraZeneca Pty Ltd, Eli Lilly Australia Pty Ltd, Janssen-Cilag Pty Ltd, Pfizer Australia Pty Ltd, and Roche Products Pty Ltd.

In this project, we used adjusted microsimulation models from Health&WealthMOD and Health&WealthMOD2030 to project the economic costs of arthritis. The development of the Health&WealthMOD was funded by the Australian Research Council (ARC) Linkage Project Grant LP0774919 and awarded to Chief Investigators: Deborah Schofield, Megan Passey, Arul Earnest, Richard Percival and Simon Kelly. Health&WealthMOD2030 was funded by the ARC Linkage Project Grant LP100100158 and awarded to Chief Investigators: Deborah Schofield, Megan Passey, Simon Kelly, Theo Vos, Robert Tanton. Pfizer Australia was a partner to both these grants.

#### Authors

Professor Deborah Schofield<sup>a,b,c,</sup> Dr Rupendra Shrestha<sup>a</sup> Dr Michelle Cunich<sup>a</sup>

a Faculty of Pharmacy, Charles Perkins Centre, The University of Sydney, NSW, 2006, Australia b Murdoch Children's Research Institute, Royal Children's Hospital, Flemington Road, Parkville Victoria, 3052, Australia c Garvan Institute of Medical Research, The University of Sydney, NSW, 2006, Australia

### Executive summary

Arthritis is a common, costly and disabling condition<sup>1</sup>. Due to its impact on physical functioning (and psychological impacts due to pain), it is a condition that impacts substantially on labour force participation (including employment, absenteeism, presenteeism). The indirect costs of arthritis (labour force participation, travel costs, special aids and equipment, carer costs) are recognised as being greater than the direct (healthcare) costs, and these costs are expected to increase over time due to demographic factors (population ageing, more older women in the population who have a greater risk of arthritis)<sup>1-3</sup>.

The aims of this study were: (a) To project the economic impacts of arthritis from 2015 to 2030, primarily focussing on the costs of lost productivity; and (b) the potential savings from implementing the main elements of an effective intervention for managing arthritis from the societal perspective. The projected costs of arthritis to the health system under these same scenarios are considered in *Counting the Cost, Part 1 Healthcare Costs.* 

In this report, we projected the costs of arthritis through lost labour force participation among Australians aged 15-64 years from the individual and government perspectives. Using our adjusted microsimulation models, we projected 52,000 people to be out of the labour force due to arthritis in 2015, increasing to 59,000 in 2030 – a 13% increase.

By 2030, people not in the labour force due to arthritis were projected to receive AU\$1,194.23 (95%CI: \$1,260.73; \$1,139.16) less in income, \$295.53 (95%CI: \$291.04; \$299.75) more in welfare payments and pay \$307.55 (95%CI: \$288.92; \$328.94) less in taxes per week. They would also have \$431,719 (95%CI: \$374,393; \$493,846) less in total savings and assets than full-time workers without arthritis.

National costs consisted of a loss in annual personal income of \$1,753 million (95%CI: \$1,631 million; \$1,911 million) in 2015, increasing to \$2,626 million (95%CI: \$2,477 million; \$2,808 million) in 2030 – a 50% increase. Extra annual welfare payments increased from \$635 million (95%CI: \$619 million; \$649 million) in 2015 to \$784 million (95%CI: \$767 million; \$801 million) in 2030 – a 23% increase. Lost annual taxation revenue increased from \$458 million (95%CI: \$422 million; \$499 million) in 2015 to \$660 million (95%CI: \$612 million; \$714 million) in 2030 – a 12% increase. A loss of \$7.2 billion in GDP was projected for 2015 due to the impact of arthritis on the labour force, increasing to \$9.4 billion in 2030 (all figures are expressed in 2015 real Australian dollars). In this report we also projected the costs of caring for a relative or friend with arthritis through lost labour force participation from 2015 to 2030.

Using our models, we estimated there were 19,000 primary carers (aged 15-64 years) not in the labour force and caring for people with arthritis in 2015. Carers of people with arthritis who were out of the labour force received \$293.53 in median weekly income in 2015, which was only 23% of the median weekly income of non-carers who are employed full-time. These primary carers received a median amount of weekly welfare payments of \$246.70 and paid a median value of \$0 in tax per week, whereas non-carers in full-time employment received a median value of \$0 in welfare payments per week, and paid a median value of \$256.09 per week in tax in 2015.

By 2030, there are projected to be 22,000 (0.12% of total population) people out of the labour force and caring for someone with arthritis. Primary carers who are not in the labour force and caring for someone with arthritis are projected to receive \$313.34 a week in income, \$247.78 a week in welfare payments, and pay \$0 in income taxes in 2030, expressed in real Australian 2015 dollars.

Published results on the health related quality of life outcomes of a randomised controlled trial The Arthritis, Diet, and Activity Promotion Trial (ADAPT)<sup>4,5</sup> were used to estimate the number of people, who would have remained in the labour force if their knee osteoarthritis (OA) had been managed with the ADAPT interventions. It has been estimated that there would be an additional 572 people in the workforce aged 15-64 years as a result of managing their knee OA through a dietary weight loss plus exercise intervention from the ADAPT trial in 2015, which would increase to 716 individuals in 2030. The cumulative economic benefit of this increased labour force participation would be an estimated increase of \$20.8 million in their annual income in 2015, which would increase to \$33.5 million in 2030. There would be an estimated saving of about \$7.8 million in welfare payments and an estimated increase of \$5.4 million in income tax revenue in 2015. By 2030, the economic benefits for the government would be an estimated reduction of \$9.9 million per year in welfare payments, and an estimated increase of \$8.4 million in income tax revenue – all attributable to the individuals being able to remain in the labour force as a result of pain reduction by managing their knee OA through a *dietary weight* loss plus exercise intervention. This demonstrates that the interventions to manage knee OA or other chronic conditions have the potential to provide economic benefits to individuals in terms of their increased labour force participation and private income, and to the government in terms of increased income tax revenue and decreased transfer payments, in addition to health benefits.

### Background

Global Burden of Disease Study Collaborators (2015) estimated there were 537.6 million years lived with disability (YLDs) globally due to acute and chronic diseases and injuries in 1990, increasing to 764.8 million in 2013 (42.3% increase) due to population growth and ageing<sup>6</sup>. Musculoskeletal disorders were a principal category of chronic disease contributing to the increase in YLDs. YLDs for rheumatoid arthritis increased by 56.8% and, for osteoarthritis by 75.4%, from 1990 to 2013. Although these studies provide information about the disease burden of arthritis, governments and international public health/economic organisations (such as WHO, OECD) are equally concerned about the impact of chronic diseases such as arthritis on the productivity of citizens and the capacity of future governments to provide healthcare and other essential services for the ageing population<sup>7-10</sup>.

Arthritis is a common condition globally. An estimated 22.7% (52.5 million) of people aged 18 years or older in the United States have doctor-diagnosed arthritis, and 43.2% also reported activity limitations due to arthritis in 2010-2012<sup>11</sup>. Prevalence of arthritis is lower in Europe, mainly due to the lower prevalence of inactivity and obesity in European women (e.g. 13% in the United Kingdom)<sup>12</sup>. Arthritis and musculoskeletal conditions have been a National Health Priority Area (NHPA) in Australia since 2002<sup>13</sup>. These conditions are more prevalent than any other NHPA conditions, affecting around 6.1 million people (26.9% of the population) in 2012<sup>1</sup>. Importantly, more than half of the population with musculoskeletal conditions in Australia (58.4%) is currently aged 25-64 years - the principal working-age population. Prevalence is estimated to increase in the next 17 years due to population ageing, with the number of cases of arthritis and other musculoskeletal conditions projected to increase by 43% to 8.7 million (increase of 2.6 million), affecting 30.2% of the population by 2032<sup>1</sup>. Arthritis also accounts for around 13% of the disability reported in Australia<sup>14</sup>.

The direct (healthcare) costs of arthritis are substantial and rising, due mainly to population ageing. In 2003, the total direct costs attributable to arthritis and other rheumatic conditions (AORC) were \$80.8 billion in the United States. An estimated 46.1 million people aged 18 years or older reported having AORC, and their average per-person direct costs were \$1,752. Ambulatory care accounted for the highest per-person direct costs (\$914), followed by emergency department and inpatient services (352), prescriptions (338), and other costs (146)<sup>15</sup>.

In Australia, arthritis and other musculoskeletal conditions are the fourth most expensive disease group in terms of health expenditure and the fourth most common reason for seeking GP and specialist services<sup>14</sup>. Healthcare expenditure allocated to these conditions increased from \$4.0 billion in 2004-0516 to \$5.7 billion in 2008-09 and it has remained the fourth most expensive disease group, accounting for 8.7% of total healthcare expenditure allocated to disease groups (\$65.1 billion) in these years<sup>2</sup>. Around 54% of healthcare expenditure on arthritis and other musculoskeletal conditions was on hospital admitted patient services (\$3.1 billion), followed by 30% on out-of-hospital medical expenses (\$1.7 billion), and 16% on prescriptions (\$922 million)<sup>2</sup>. Recent estimates of the direct costs for treating arthritis and other musculoskeletal conditions are even higher at \$9.2 billion in 2012<sup>1</sup>.

Counting the Cost Part 1 Healthcare Costs (published separately) estimates that the cost of arthritis alone to the Australian health system was more than \$5.5 billion in 2015, and that this would increase to more than \$7.6 billion by 2030.

However, the full-range of indirect costs (i.e. lost earnings, lost productivity, lost superannuation, lost taxation revenue, carer costs, extra welfare payments, cost of aids and home modifications, travel costs) are considered to be even greater than the direct costs<sup>1,17</sup>. These costs are mostly attributed to lost productivity, with arthritis affecting an individual's ability to maintain employment (and earnings) due to pain and physical disability<sup>1,18</sup>.

In Australia in 2009, the estimated cost of arthritis through lost labour force participation among older workers (45-64 years) consisted of lost income of AU\$3,787 million, extra welfare payments of AU\$291 million, and lost taxation revenue of AU\$394 million<sup>19</sup>. The impact of arthritis on labour force participation among older workers has continued to be significant and enduring in Australia. A recent study has shown that 13.3% of people aged 45-64 years out of the labour force due to ill-health exited because of their arthritis in 2010 (45,000 people) with a similar proportion projected to exit in 2030 (60,000 people).

## 1 Background

In this and other studies, arthritis is consistently the second most common chronic condition (after back problems) leading to exits from the labour force for this age group<sup>20</sup>. Workforce absences form a significant proportion of the burden of the disease in many countries, including the United Kingdom, Canada, and the United States<sup>3,21</sup>.

Arthritis is also the second most common disabling condition among care recipients who have informal carers who are not in the labour force in Australia<sup>22</sup>, and thus the disease significantly impacts on the productivity of carers as well.

There are few studies on the costs of arthritis through lost labour force participation that consider the societal perspective<sup>17,20</sup>. Most studies on the indirect costs of arthritis (productivity) focus only on lost earnings to individuals. Ensuing extra welfare payments and reduced income taxation revenue impacting on governments are rarely assessed<sup>19</sup>. Although the authors have estimated some of the economic costs of arthritis through its impact on labour force of older workers (aged 45-64 years) in previous studies<sup>19,23</sup>, they are limited to a single year (2009). Additionally, there is limited research on the costs of caring for a relative or friend with arthritis<sup>1</sup>.

#### 1.1 Objectives

The objectives of this study are:

- to project the economic impacts of arthritis up to year 2030, primarily through lost productivity due to the disease;
- (2) to examine the potential economic benefits as a result of increased productivity from implementing key elements of the ADAPT interventions<sup>4,5,24</sup>.

Projections of the health system costs associated with arthritis to 2030 are considered in *Counting the Cost Part 1 Healthcare Costs*.

The study of the economic impacts consisted of two stages.

#### Stage 1

The aim for Stage 1 was to project multiple costs of arthritis among Australians aged 15-64 years (lost labour force participation, lost income, extra welfare payments, lost taxes, lost GDP, and carer costs) over a 15 year period (2015-30) using our microsimulation models of the impacts of ill-health on individuals, government and society as a whole.

We developed a microsimulation model for lost productivity due to arthritis and the associated economic impacts. The model was used to project the following economic outcomes, for every five years from 2015 to 2030, using the 'business as usual scenario' to quantify:

- the number of people out of the labour force due to arthritis
- the number of people working full-time and part-time with arthritis
- the loss of income and wealth (including superannuation) for people out of the labour force due to arthritis, relative to those who were able to continue to work full-time without arthritis
- the lost taxation revenue and increased government welfare costs due to arthritis
- the lost GDP due to people being out of employment because of arthritis
- the costs of lost productivity, including loss of income to individuals and loss of taxation revenue and increased welfare payments to the government) of becoming an informal carer for someone with arthritis.

The model captured the impacts of demographic changes including population ageing and changing labour force patterns (such as the increasing number of women and older workers in the labour force and real wages growth) from 2015 to 2030.

#### Stage 2

Health interventions for people with arthritis can prevent or slow disease progression, reduce pain and immobility caused by arthritis, preserve independence, and improve quality of life<sup>4,25</sup>. Reducing pain and disease progression in people with arthritis potentially enables them to continue in the labour force and contribute to the economy. The *Arthritis, Diet, and Physical Activity Promotion Trial* (ADAPT) – a single-blind, randomised controlled trial (RCT) involving older, obese patients (age>=60 years) with knee osteoarthritis (OA) – assessed strategies for improving quality of life (bodily pain) in

## Background

people with knee OA<sup>4,5,24</sup>. In this trial, participants were assigned to one of four treatments: exercise only, dietary weight loss only, dietary weight loss plus exercise, or a healthy lifestyle control condition with 18 months of follow-up. The aim of Stage 2 was to estimate the potential productivity benefits associated with implementing *ADAPT* interventions in the Australian population with knee OA. An evaluation of the cost-effectiveness of implementing these interventions in the Australian health system is included in *Counting the Cost Part 1 Healthcare Costs*.

We estimated the potential labour force benefits of managing knee OA by implementing key elements of the ADAPT study, specifically dietary weight loss only, exercise only or dietary weight loss plus exercise programs for people with knee OA<sup>4,5,24</sup>. We also estimated the flow-on economic benefits associated with the potential increase in labour force participation, including increased income to individuals, and increased income taxation revenue and reduced welfare payments to the government.

### 1.2 Overview of model development

#### Stage 1

We adjusted our current microsimulation models, Health&WealthMOD and Health&WealthMOD2030<sup>26,27</sup>, to develop a model to project the economic impacts of lost labour force participation due to arthritis. Microsimulation is one of the most robust economic applications in Australia. It captures data on a large number of individuals and is used in the development of policy at the Commonwealth Treasury, Department of Human Services and other large social policy departments. Our current models only focus on older workers (aged 45-64 years). For this study, the model was extended to all individuals aged 15-64 years.

#### Stage 2

For Stage 2, we used the effect size estimates of managing knee OA by implementing the key elements of the *ADAPT* trial<sup>4,5,24</sup> on labour force participation and other economic outcomes from published literature where available.

In the absence of direct measures of the effect on labour force participation in the published literature, we modelled improvements in labour force participation rates by analysing the effects of managing knee OA by implementing the key elements of ADAPT strategies on individuals' quality of life/health status (such as the SF-36 bodily pain domain score)<sup>28</sup>. The Household, Income and Labour Dynamics in Australia (HILDA) Survey data<sup>29</sup> was used to model the association between SF-36 and labour force participation. The potential increase in labour force participation rates associated with the improved SF-36 scores of the individuals with knee OA from implementing ADAPT interventions were then estimated. We then simulated the national impact on the labour force in terms of the number of people with knee OA who would have remained in the labour force as a result of implementing ADAPT interventions, who would otherwise be out of the labour force. We also estimated the economic benefits associated with the potential increase in the labour force.

#### 1.3 Methodology

Our microsimulation models, Health&WealthMOD and Health&WealthMOD2030, were adjusted to project the costs of arthritis through lost labour force participation to individuals, government and society from 2015 to 2030. The development of our latest model, Health&WealthMOD2030, is described in Schofield et al (2014)<sup>27</sup>.

The primary data for this project came from the Surveys of Disability, Ageing and Carers (SDAC) 2003, 2009 and 2012<sup>30-32</sup>. The SDACs 2003, 2009 and 2012 are nationally representative (large) Australian household surveys conducted by the Australian Bureau of Statistics (ABS). They provide the most comprehensive data on individuals in terms of their personal characteristics (such as age, sex, family type), socioeconomic status (such as education, income, labour force participation, employment restrictions, reasons for retirement, home ownership, receipt of welfare payments), chronic conditions and disabilities (main chronic condition/ disability), and nature of any informal care required as well as information on their carers<sup>30-32</sup>.

# 1 Background

Although the SDACs 2003, 2009 and 2012 are a rich source of information on people with a chronic condition (such as arthritis) and their informal care needs, they are limited in terms of economic data. For more detailed economic information (such as income, welfare payments, income taxes paid and the value of different types of wealth assets e.g. superannuation), the outputs from another microsimulation model, Static Incomes Model or STINMOD, developed by the National Centre for Social and Economic Modelling (NATSEM) for the Australian Government<sup>33</sup> were used. STINMOD is Australia's leading static microsimulation model<sup>34</sup> and routinely used by the Commonwealth departments for assessing distributional and revenue implications of tax and cash transfer reforms. The model operates at the micro-level of individuals and families, and is based on the ABS' Income Distribution Survey unit record files<sup>35</sup>. This model simulates the impact of major federal government cash transfers, income tax and the Medicare levy on individuals and families in Australia.

To account for demographic changes in the Australian population from 2015 to 2030 (such as population ageing and trends in labour force participation) we used the population and the labour force projections for this time period provided by the Commonwealth Treasury that were used in the 2015 Intergenerational Report. For socio-economic changes in the population, we used outputs from another microsimulation model, Australian Population and Policy Simulation Model or APPSIM, which was also developed by NATSEM in collaboration with 12 Australian Government departments. It is a dynamic population microsimulation model that was developed to provide a snapshot of the socio-economic characteristics of the Australian population annually<sup>36</sup>. The model uses a sample of 188,000 records (1% of the 2001 Australian census population) to simulate the impact of social and fiscal policies on every Australian.

The SDACs 2003, 2009 and 2012 were reweighted separately using the ABS reweighting algorithm GREGWT<sup>37</sup> to account for demographic and other changes in the population occurring between survey years (2003, 2009, 2012) ensuring that, together they represented the Australian population in 2015, and projected to 2020, 2025, and 2030. Commonwealth Treasury provided demographic (ageing and labour force) projections for 2015 to 2030.

The economic data from STINMOD for 2015 were indexed to reflect economic growth projections from 2015 to 2030. Income and taxes paid by individuals were assumed to grow at a rate of 1% per annum in real terms (Treasury, 2015)<sup>38</sup>. Welfare payments were assumed to have no real growth based on the government's policy of increasing welfare payments (except for aged pension) in line with national Consumer Price Index (CPI) growth.

#### 1.4 Structure of the report

Section 2 estimates the economic costs of arthritis through lost productivity for individuals, government and society from 2015 to 2030. It includes a description of the cost items measured and estimated (as per Stage 1), and the projections of these costs. Section 3 examines the costs of arthritis through lost productivity due to informal caregiving from 2015 to 2030. It includes projections of the costs of lost labour force participation due to people (aged 15-64 years) taking on the role of informal carer for someone with arthritis (measured in terms of income, welfare payments, taxes paid and savings differences compared to working non-carers). Section 4 provides discussion and concluding remarks about these costs of arthritis. Section 5 provides estimates of the gains in productivity if people with arthritis could be engaged in the ADAPT trial strategies. Section 6 provides an overall conclusion.

## The economic costs of arthritis

through lost productivity on individuals, government and society from 2015 to 2030

Using our adjusted models, we estimated the costs of lost labour force participation due to arthritis among Australians aged 15-64 years from 2015 to 2030. The following are the definitions of the key economic outcomes analysed.

## 2.1 Lost labour force participation due to arthritis

Respondents to the Survey of Disability, Ageing and Carers were asked to nominate their current labour force status. Those who were not in the labour force were also asked to nominate the main reason they were not working or looking for work e.g. 'own ill-health or disability'.

The surveys collected information from all respondents on the type of long term health conditions they have and respondents were also asked to nominate their main long term health condition among the conditions they have. "Arthritis and related disorders" (ICD10 code M00-19) is one of the conditions on the list and respondents who self-reported having this health condition were considered to have 'arthritis'.

Respondents identified as (a) being out of the labour force due to their own ill-health or disability, and (b) having arthritis as their main condition were considered to have *lost labour force participation due to arthritis*.

#### 2.2 The economic costs

The following costs of arthritis through lost labour force participation are projected for every five years from 2015 to 2030:

- the number of people out of the labour force due to arthritis
- the number of people working full-time and parttime with arthritis
- the loss of income and wealth (including cash, assets and superannuation) for people out of the labour force due to arthritis relative to those who are able to work full-time and do not have arthritis and relative to those who are able to work part-time and do not have arthritis

- the lost taxation revenue and increased welfare payments to government due to the lost labour participation because of arthritis
- the GDP loss attributable to arthritis through lost labour force participation among people aged 15-64 years.

#### 2.3 Measurement of economic costs

Personal income consists of labour market earnings, income from other sources generating a return (such as rental properties, investments, interest on cash in a bank), and welfare payments.

Relevant welfare payments for this age group consists of Disability Support Pension, Newstart Allowance (for people looking for work), Youth Allowance, Carer Payment, and Family Tax Benefit; see http://www. humanservices.gov.au/customer/services.

The taxes paid by individuals included personal income tax and the Medicare levy.

Total wealth consists of cash savings, the value of owner occupied home, investment properties, shares, superannuation balance and other investments. Information on wealth variables are only available for income units (e.g. for couples or families) and are the total for an income unit. The value of wealth for an individual, including savings, assets and superannuation, was calculated by dividing the total value of wealth of an income unit by the number of adults (15 years and over) in that income unit and thus, everyone in an income unit was considered to have an equal value of personal wealth.

We calculate the impact of arthritis on national Gross Domestic Product (GDP) using the Commonwealth Treasury's GDP formula:

 $GDP = (GDP/H) \times (H/EMP) \times (EMP/LF) \times (LF/Pop15+) \times Pop15+$ 

where GDP = Gross Domestic Product; H = total hours worked; EMP = total number of persons employed; LF = total labour force; and Pop15+ = population aged 15 years and over<sup>39</sup>.

### The economic costs of arthritis

through lost productivity on individuals, government and society from 2015 to 2030

#### 2.4 Statistical simulation

Descriptive analysis is undertaken to establish patterns in projected lost labour participation due to arthritis and the economic costs listed above in 2015, 2020, 2025, and 2030. We present the mean, standard deviation and median weekly income and savings, welfare payments, and taxes paid by people aged 15-64 years who are out of the labour force due to arthritis or in the labour force (employed full-time with and without arthritis, employed part-time with and without arthritis). All figures are expressed in real 2015 Australian dollars.

The differences in economic outcomes of those not in the labour force due to arthritis compared to those in full-time or part-time employment who do not have arthritis, were estimated using counterfactual simulation with Monte Carlo methods. For each record of those not in the labour force due to arthritis, a counterfactual record was selected at random with replacement from the pool of those in full-time employment who do not have arthritis; records were matched for age group, sex and highest level of education. The mean of the difference in the economic outcomes between the records of those not in the labour force due to arthritis and their counterfactuals were estimated. We ran 1,000 simulations, generating 1000 counterfactual datasets for records of those not in the labour force due to arthritis. The average of the 1000 simulations and the 95% uncertainty interval, estimated using the percentile method, are reported as the result in this report.

The simulation method was repeated with the selection of counterfactual records from the pool of those in part-time employment without arthritis to estimate the differences in economic outcomes of those not in the labour force due to arthritis and those in part-time employment who do not have arthritis.

For the estimation of the economic costs of lost labour force participation due to arthritis at the national level, the counterfactuals were drawn from the pool of those in the labour force (i.e. employed full-time or employed part-time or unemployed) who do not have arthritis.

#### 2.5 Results

Among approximately 15.8 million people aged 15-64 years in 2015, about 52,000 (0.33%) were out of the labour force due to arthritis; about 354,000 (2.25%) were employed full-time with arthritis; about 193,000 (1.23%) were employed part-time with arthritis; about 7.6 million (48.58%) were employed full-time without arthritis; and about 3.3 million (20.88%) were employed part-time without arthritis, 35.3% were working part-time compared to 30% of those without arthritis (Table 1).

Those who were out of the labour force due to arthritis received \$333.13 in median weekly income in 2015, which is only about one-fourth of the median weekly income of those employed full-time without arthritis (\$1,287.81) (Table 1). Those not in the labour force due to arthritis received a median value of weekly welfare payments of \$329.50, whereas those in full-time employment received a median value of \$0 welfare payments per week and those in part-time employment received a median value of the labour force due to arthritis paid a median value of \$0 in tax per week whereas those employed full-time, who do not have arthritis, paid a median of \$256.21 per week in tax in 2015.

An estimated median value of total wealth, including savings, assets and superannuation of those not in the labour force due to arthritis was about \$156,000. This is about \$200,000 less than the wealth of those in full-time employment who do not have arthritis and who had an estimated median value of total wealth of about \$345,000. Those who are working part-time and do not have arthritis have a median value of total wealth of about \$321,000 in 2015 (Table 1).

Table 1: Mean and median weekly income, welfare payments, taxes and total value of savings and assets of individuals with and without arthritis, Australian population aged 15-64 years (in 2015 Australian dollars)

	z		2015	2			2020	0			2025	2			2030	0	
Labour force status^	Survey records	Weighted population (%)	Mean	SD	Median	Weighted population (%)	Mean	S	Median	Weighted population (%)	Mean	S	Median	Weighted population (%)	Mean	SD	Median
	-					We	Weekly total income (AU\$) of individuals	income (A	VU\$) of inc	lividuals					-	-	
Employed full-time without arthritis	53,949	7,644,000 (48.6%)	1,545.53	1,405.97	1,287.81	8,290,000 (49.5%)	1,627.80	1,490.30	1,351.23	8,871,000 (50.6%)	1,729.73	1,598.68	1,421.27	9,381,000 (50.3%)	1,858.54	1,691.18	1,513.72
Employed full-time with arthritis	2,497	354,000 (2.3%)	1,532.98	1,237.24	1,316.40	395,000 (2.4%)	1,598.78	1,317.50	1,375.37	425,000 (2.4%)	1,727.33	1,411.50	1,452.87	452,000 (2.4%)	1,856.19	1,524.46	1,542.24
Employed part-time without arthritis	23,707	3,285,000 (20.9%)	664.61	1,169.21	492.20	3,316,000 (19.8%)	732.36	1,284.09	552.42	3,561,000 (20.3%)	769.76	1,396.89	576.41	3,780,000 (20.3%)	816.19	1,509.69	621.77
Employed part-time with arthritis	1,471	193,000 (1.2%)	697.39	701.93	605.43	214,000 (1.3%)	724.02	752.21	613.97	230,000 (1.3%)	791.96	812.74	671.18	242,000 (1.3%)	859.70	853.52	720.86
Not in labour force due to arthritis	348	52,000 (0.3%)	310.87	205.52	333.13	55,000 (0.3%)	315.22	216.00	334.84	57,000 (0.3%)	329.02	220.39	356.63	59,000 (0.3%)	343.51	227.86	368.93
						Weekly v	/elfare inco	ome (AU\$	) received	Weekly welfare income (AU\$) received by individuals	ıals						
Employed full-time without arthritis	53,949	7,644,000 (48.6%)	26.52	82.07	-0.0	8,290,000 (49.5%)	26.18	81.11	0.0	8,871,000 (50.6%)	27.36	83.07	0.0	9,381,000 (50.3%)	27.20	83.62	0.0
Employed full-time with arthritis	2,497	354,000 (2.3%)	25.36	79.56	0.0	395,000 (2.4%)	25.41	79.45	0.0	425,000 (2.4%)	25.91	80.38	0.0	452,000 (2.4%)	26.32	81.68	0.0
Employed part-time without arthritis	23,707	3,285,000 (20.9%)	84.31	145.40	5.75	3,316,000 (19.8%)	85.45	145.44	5.75	3,561,000 (20.3%)	85.42	144.57	5.75	3,780,000 (20.3%)	85.31	144.80	5.75
Employed part-time with arthritis	1,471	193,000 (1.2%)	88.80	157.42	5.75	214,000 (1.3%)	86.16	155.00	5.75	230,000 (1.3%)	87.88	156.52	0.0	242,000 (1.3%)	89.67	157.85	0.0
Not in labour force due to arthritis	348	52,000 (0.3%)	281.63	200.25	329.50	55,000 (0.3%)	280.63	198.58	329.50	57,000 (0.3%)	291.12	197.77	329.50	59,000 (0.3%)	299.36	196.91	329.50

Table 1: Mean and median weekly income, welfare payments, taxes and total value of savings and assets of individuals with and without arthritis, Australian population aged 15-64 years (in 2015 Australian dollars) continued

	z		2015	15			2020	20			2025	25			20	2030	
Labour force status^	Survey records	Weighted population (%)	Mean	ß	Median	Weighted population (%)	Mean	SD	Median	Weighted population (%)	Mean	S	Median	Weighted population (%)	Mean	S	Median
						Weekly tax p	paid (inclu	ides Medic	care levy)	tax paid (includes Medicare levy) (AU\$) by individuals	dividuals						
Employed full-time without arthritis	53,949	7,644,000 (48.6%)	357.72	497.77	256.21	8,290,000 (49.5%)	379.06	538.33	268.91	8,871,000 (50.6%)	406.21	584.84	283.02	9,381,000 (50.3%)	440.57	631.40	302.76
Employed full-time with arthritis	2,497	354,000 (2.3%)	347.83	438.72	254.10	395,000 (2.4%)	363.38	466.62	258.98	425,000 (2.4%)	395.11	504.89	276.16	452,000 (2.4%)	428.11	550.24	299.82
Employed part-time without arthritis	23,707	3,285,000 (20.9%)	77.19	474.51	4.41	3,316,000 (19.8%)	88.80	525.03	10.05	3,561,000 (20.3%)	94.03	578.23	10.07	3,780,000 (20.3%)	100.77	628.95	14.77
Employed part-time with arthritis	1,471	193,000 (1.2%)	75.47	188.98	15.84	214,000 (1.3%)	80.69	203.28	15.12	230,000 (1.3%)	00.06	216.70	19.87	242,000 (1.3%)	99.54	229.92	26.10
Not in labour force due to arthritis	348	52,000 (0.3%)	0.88	22.10	0.0	55,000 (0.3%)	1.80	29.57	0.0	57,000 (0.3%)	2.25	32.73	0.0	59,000 (0.3%)	2.98	37.34	0.0
					Total value	Total value of savings and assets including superannuation (AU\$) of individuals^ $^{\Lambda}$	and asset:	s including	g superan	nuation (AU	\$) of indi	viduals^^					
Employed full-time without arthritis	53,949	7,644,000 (48.6%)	497,239	695,920	345,317	8,290,000 (49.5%)	579,372	817,885	401,515	8,871,000 (50.6%)	689,349	985,677	478,904	9,381,000 (50.3%)	814,417	1,172,728	563,307
Employed full-time with arthritis	2,497	354,000 (2.3%)	589,323	726,800	423,645	395,000 (2.4%)	697,503	883,014	496,537	425,000 (2.4%)	836,362	1,066,788	591,149	452,000 (2.4%)	999,921	1,294,570	700,649
Employed part-time without arthritis	23,707	3,285,000 (20.9%)	468,691	729,426	321,381	3,316,000 (19.8%)	558,377	826,061	382,767	3,561,000 (20.3%)	651,503	882,114	445,557	3,780,000 (20.3%)	761,746	957,262	519,902
Employed part-time with arthritis	1,471	193,000 (1.2%)	534,189	690,894	344,462	214,000 (1.3%)	626,518	826,826	393,961	230,000 (1.3%)	744,165	993,049	468,943	242,000 (1.3%)	881,321	1,178,348	554,308
Not in labour force due to arthritis	348	52,000 (0.3%)	333,658	618,429	156,231	55,000 (0.3%)	385,820	720,790	180,204	57,000 (0.3%)	437,942	844,626	201,014	59,000 (0.3%)	503,943	991,740	207,377
Note:																	

### The economic costs of arthritis

through lost productivity on individuals, government and society from 2015 to 2030

By 2030, the working-age population is projected to be 18.6 million and consist of 59,000 people who were out of the labour force due to arthritis; 452,000 employed full-time with arthritis; 242,000 employed part-time with arthritis; 9.4 million employed full-time without arthritis; and 3.8 million employed part-time without arthritis. Those with lost labour force participation due to arthritis are projected to receive \$368.93 a week as their total income, \$329.50 a week in welfare payments, and pay \$0 in income taxes in 2030, expressed in real 2015 dollars (Table 1, last column).

Compared to those in full-time employment without arthritis, people out of the labour force due to arthritis receive \$957.04 (95%CI: \$907.32; \$1,018.73) less per week in total income after adjusting for age, sex and highest level of education in 2015 (Table 2). They also receive significantly more in welfare payments (an extra AU\$280.45 per week, 95%CI: \$275.61; \$285.16) and pay significantly less in taxation (\$252.05 per week, 95% CI: \$236.55; \$270.37) compared to those working full-time without arthritis. When compared to those in part-time employment without arthritis, those not in the labour force due to arthritis receive \$341.38 (95% CI: \$303.92; \$383.22) less per week in total income and pay \$75.80 (95% CI: \$65.46; \$88.85) less in tax per week but receive an extra \$236.51 (95% CI: 228.05; \$244.45) per week in welfare payments in 2015. The mean differences in weekly income, welfare payments and taxation between those with lost labour force participation due to arthritis and employed full-time without arthritis and employed part-time without arthritis were also estimated for 2030 (Table 2, last two columns). Lost income as a result of being out of the labour force due to arthritis is projected to increase from \$957.04 per week in 2015 (95%CI: \$907.32; \$1,018.73) to \$1,194.23 per week in 2030 (95% CI: \$1,139.16; \$1,260.73) in real terms (compared to those employed full-time without arthritis). People with lost labour force participation due to arthritis are estimated to pay \$252.05 (95% CI: \$236.55; \$270.37) per week less in income taxes than those employed full-time without arthritis in 2015, with the difference increasing to \$307.55 per week (95% CI: \$288.92; \$328.94) in 2030

For 2015, the total value of wealth including savings, assets and superannuation of those not in the labour force due to arthritis is about \$263,000 (95%CI: \$224,000; \$309,000) less than those with no arthritis and in full-time employment. This is about \$200,000 (95%CI: \$158,000; \$242,000) less than the total value of wealth of those who do not have arthritis who are in part-time employment. The differences in the total value of wealth between those not in the labour force due to arthritis and those in full-time employment with no arthritis, adjusted for age, sex and highest level of education, is projected to increase to about \$432,000 (95%CI: \$374,000; \$494,000) in 2030. Between those not in the labour force due to arthritis and those in part-time employment with no arthritis, this difference is projected to increase to about \$320,000 (95% CI: \$263,000; \$388,000) in 2030.

The national economic impacts of arthritis through lost labour force participation consists of \$1,753 million (95%CI: \$1,631; \$1,911 million) in lost income in 2015, increasing to \$2,626 million (95%CI: \$2,477; \$2,808 million) in 2030 (i.e. 61% increase over 15 years) mainly due to population growth and ageing (Table 3). Additional welfare payments because of lost labour force participation due to arthritis are projected to increase by about 23% over this period, from \$635 million (95%CI: \$619; \$649 million) in 2015 to \$784 million (95%CI: \$767; \$801 million) in 2030. Finally, lost annual taxation revenue is projected to increase by about 44% in real terms, from \$458 million (95%CI: \$422; \$499 million) in 2015 to \$660 million (95%CI: \$612; 714 million) in 2030 (Table 3).

As a result of workers aged 15-64 years withdrawing from the labour market because of their arthritis, there are also significant losses in GDP each year. Using the Commonwealth Treasury's GDP formula, we calculated these losses to be \$7.2 billion, \$7.9 billion, \$8.6 billion, and \$9.4 billion in lost GDP in 2015, 2020, 2025 and 2030, respectively. Table 2: Mean differences in weekly income, welfare payments, taxes and total value of savings and assets between people with lost productivity due to arthritis and those employed full-time without arthritis and those employed part-time without arthritis (adjusted for age, sex and highest level of education), Australian population aged 15-64 years (in 2015 Australian dollars)

	2	2015		2020		2025		2030
Labour force status	<pre>\$ difference</pre>	95% CI	<pre>\$ difference</pre>	95 % CI	<pre>\$ difference</pre>	95% CI	<pre>\$ difference</pre>	95% CI
			Weekly total inco	Weekly total income (AU\$) of individuals	lls			
Not in labour force due to arthritis compared to employed full-time without arthritis	-957.04	(-1,018.73; -907.32)	-1,018.11	(-1,085.19; -967.04)	-1,105.64	(-1,162.58; -1,051.26)	-1,194.23	(-1,260.73; -1,139.16)
Not in labour force due to arthritis compared to employed part-time without arthritis	-341.38	(-383.22; -303.92)	-360.38	(-402.32; -322.32)	-409.72	(-456.80; -371.40)	-453.15	(-495.27; -412.50)
		Week	dy welfare income	Weekly welfare income (AU\$) received by individuals	lividuals			
Not in labour force due to arthritis compared to employed full-time without arthritis	280.45	(275.61; 285.16)	277.51	(273.11; 282.25)	287.28	(282.67; 291.74)	295.53	(291.04; 299.75)
Not in labour force due to arthritis compared to employed part-time without arthritis	236.51	(228.05; 244.45)	234.06	(225.72; 241.94)	243.03	(234.74; 250.89)	250.54	(242.23; 258.08)
		Weekly ta	ax paid (includes N	Weekly tax paid (includes Medicare levy) (AU\$) by individuals	y individuals			
Not in labour force due to arthritis compared to employed full-time without arthritis	-252.05	(-270.37; -236.55)	-265.55	(-285.10; -248.42)	-286.50	(-305.80; -268.46)	-307.55	(-328.94; -288.92)
Not in labour force due to arthritis compared to employed part-time without arthritis	-75.80	(-88.85; -65.46)	-79.04	(-91.08; -68.42)	-87.17	(-100.22; -76.78)	-92.53	(-104.55; -80.98)
		Total value of savin	igs and assets incl	$\mathfrak{g}\mathfrak{s}$ and assets including superannuation (AU\$) of individuals^	(AU\$) of individ	vals^		
Not in labour force due to arthritis compared to employed full-time without arthritis	-263,327	(-309,088; -224,509)	-295,595	(-348,233; -251,908)	-361,331	(-424,036; -306,129)	-431,719	(-493,846; -374,393)
Not in labour force due to arthritis compared to employed part-time without arthritis	-193,919	(-241,842; -158,460)	-217,458	(-273,612; -178,048)	-263,864	(-320,261; -219,697)	-320,147	(-388,378; -263,457)

Table 3: National economic costs of lost labour force participation due to arthritis per year, Australian population aged 15-64 years (in 2015 Australian dollars, in millions)

	2015	15	2020	20	20	2025	20	2030
Cost	\$ impact	95% CI						
Lost income	1,753	(1,631; 1,911)	2,034	(1,900; 2,197)	2,336	(2,190; 2,503)	2,626	(2,477; 2,808)
Extra welfare payments	635	(619; 649)	681	(664; 698)	734	(716; 751)	784	(767; 801)
Lost tax revenue	458	(422; 499)	526	(482; 572)	596	(549; 647)	660	(612; 714)

#### The costs of arthritis through lost productivity due to informal caregiving from 2015 to 2030

The number of children with a chronic condition or disability and older people needing care is projected to increase substantially in the next 20 years<sup>40-42</sup>. For example, the number of older people with care needs is projected to increase almost threefold from 101 million in 2010 to 277 million in 2050 worldwide<sup>43</sup>. This care can be provided in two ways: (a) formally, by people in paid work in the healthcare sector, and (b) informally, by friends or relative of people needing care<sup>44</sup>. Informal carers normally do not receive an income for their capacity to maintain paid work<sup>45</sup>. In this section, we examine the impact of arthritis in care recipients on the labour force participation of informal caregivers.

Informal carers of people with a chronic condition or disability constituted around 12% of the Australian population (or 2.6 million people) in 2009<sup>46</sup>, and this number is expected to increase due to greater demands for such care. In adults, greater demand is largely due to Australia's ageing population<sup>40</sup> and increasing (prevalence) rates of several chronic conditions<sup>47</sup>, such as diabetes and dementia<sup>48</sup>. In children, the reasons for greater demand are complex. There has been an increase in the rate of preterm births (babies born less than 37 gestational weeks)49,50 and an increase in the survival rates of all preterm babies, including extremely preterm babies (born less than 28 gestational weeks), with an associated increase in disability amongst survivors<sup>51</sup>. The substantial number of Australians currently (and projected) providing informal care has serious ramifications for the national workforce, the tax base and welfare costs (and thus the budget balance), and GDP in addition to the individual social, emotional wellbeing, healthcare costs and other financial implications for carers.

Recognising the current (and projected) demand for informal carers worldwide, a number of studies have examined the impact of informal caregiving on labour force participation. These studies have found that the labour force participation rate for carers is lower than for non-carers<sup>45,52-55</sup>. A recent cross-sectional study involving Australians aged 45 and older (n = 265,515) demonstrated that full-time carers are not only more likely to be out of the labour force than non-carers (and part-time carers) but also more likely to have lower household income than non-carers. Just over 40% of non-carers had an annual household income of more than \$70,000, whereas only 12.6% of full-time carers had the same level of income<sup>56</sup>. Another study, using one wave of the Household Income and Labour Dynamics in Australia (HILDA) survey data (2008)<sup>57</sup>, confirmed the findings of these cross-sectional studies, by reporting that becoming a main carer reduced the probability of being in employment by around 12 percentage points for both males and females regardless of whether or not the carer lived with the care recipient. Longitudinal studies, such as Bittman et al. (2007), who examined the effects of informal caregiving on carers' employment, income and earnings using four waves of HILDA (2001–04)<sup>58</sup>, also confirmed that carers are more likely to reduce their hours of work or exit from the labour force and earn less on average than non-carers.

Few studies, however, have examined the impact of specific chronic conditions of care recipients on the labour market behaviour of carers. Schofield et al (2014), using the SDAC 2009, examined the impact of chronic conditions of care recipients on the labour force participation of carers (adjusting for age, sex, education, health of carers), as well as generating a ranking of these conditions in terms of having the greatest impact on the labour force participation of carers using logistic regression analysis<sup>22</sup>. From the SDAC 2009, they identified 1,268 respondents who were primary carers for a care recipient living with them and, of these, 628 (49.5%) were out of the labour force. The most common diseases of care recipients were: back problems (12%); arthritis and related disorders (10%); diseases of the nervous system (such as multiple sclerosis, epilepsy, cerebral palsy) (7.4%); and conditions originating in the perinatal period or congenital malformations, deformations and chromosomal abnormalities (5.1%). Whilst this study provides insight into the importance of arthritis as a common chronic condition of care recipients and affecting the earning capacity of carers, research on the economic costs associated with this is limited. Additionally, there is little on the economic consequences of medical/health interventions that not only improve the health outcomes of people with arthritis but also possibly improve the income of carers and government finances<sup>47</sup>. The aim of Section 3 is to project the costs of lost productivity (loss of incomes to individuals and loss of taxation revenue and increase in welfare payments to the government) of informal carers of people with arthritis.

## The costs of arthritis

through lost productivity due to informal caregiving from 2015 to 2030

#### 3.1 Methods

The data (and cost measures) described in Section 1.3 was used to model the costs of informal caregiving for arthritis through lost productivity out to 2030.

### Lost productivity due to being a carer of someone with arthritis

Several types of informal carers can be identified in the SDACs 2003, 2009 and 2012, such as "primary carers", "non-primary carers" and "non-resident" versus "resident carers". In this report, carers were identified as those who indicated they were:

- a) a primary carer of a usual resident (i.e. lives with) care recipient
- b) other primary carer of a usual resident care recipient;
- c) primary carer of a non-usual resident (i.e. does not live with) care recipient; or
- d) other carer.

However, information on both the main care recipient and carer (such as age, sex, education, labour force participation, chronic conditions) required for this project were only available for main usual resident care recipients matched to primary carers. For this reason, the analysis focuses only on primary carers of care recipients who live with them (i.e. categories (a) and (b) above).

This definition is consistent with the common definition of primary carers used by the ABS (2010):

A primary carer is a person who provides the most informal assistance, in terms of help or supervision, to a person with one or more disabilities or aged 60 years and over. The assistance has to be ongoing, or likely to be ongoing, for at least six months and be provided for one or more of the core activities (communication, mobility and self-care) (p. 34).

All SDAC respondents who could be identified as a primary carer (using the definition above) and reported caring for someone who had "arthritis and related disorders" (ICD10 code M00-19) as their main disabling condition were considered to be 'a primary carer caring for care recipients with arthritis' in this project. All primary carers identified as (a) being out of the labour force and (b) caring for a (live with) care recipient with arthritis as their main disabling condition were considered to have lost labour force participation due to caring for someone with arthritis.

#### 3.2 Results

Among approximately 15.8 million people aged 15-64 years, after excluding those living in caredaccommodation, in 2015, approximately 41,000 (0.26%) were primary carers of a care recipient with arthritis (living with them). Of these, 19,000 (46%) were out of the labour force; 14,000 (33%) were working full-time; and 7,800 (19%) were working part-time in 2015.

Primary carers who were out of the labour force and caring for someone with arthritis received \$293.53 in median weekly income in 2015, which is only 23% of the median weekly income of non-carers who are employed full-time (Table 4). These primary carers received a median amount of weekly welfare payments of \$246.70 and paid a median value of \$0 in tax per week, whereas non-carers in full-time employment received a median value of \$0 in welfare payments per week, and paid a median value of \$256.09 per week in tax in 2015.

By 2030, there are projected to be 22,000 (0.12% of total population) people out of the labour force and caring for someone with arthritis. Primary carers not in the labour force and caring for someone with arthritis are projected to receive \$313.34 a week in income (only 21% of the median weekly income of non-carers who are employed full-time), \$247.78 a week in welfare payments, and pay \$0 in income taxes in 2030, expressed in real 2015 dollars (Table 4, last column). The relatively lower median income of carers compared to the median weekly income of non-carers who are employed full-time is due to wages growth above that of certain government payments and income from the increasing savings of those who are able to keep working.

Table 4: Mean and median weekly income, welfare payments, taxes of primary carers of people with arthritis as main disabling condition, Australian population aged

	z		2015	2			2020	0			2025	5			2030	õ	
Labour force status^	Survey records	Weighted population (%)	Mean	SD	Median	Weighted population (%)	Mean	SD	Median	Weighted population (%)	Mean	SD	Median	Weighted population (%)	Mean	ß	Median
						Weekly	total inco	me (AU\$)	Weekly total income (AU\$) of primary carers	carers							
Employed full-time, non-carers	51,455	7,362,000 (46.8%)	1,542.60	1,406.99	1,287.47	8,002,000 (47.8%)	1,624.03	1,491.06	1,348.23	8,549,000 (48.8%)	1,726.65	1,599.87	1,420.16	9,028,000 (48.4%)	1,855.15	1,690.27	1,512.77
Employed part-time, non-carers	22,120	3,116,000 (19.8%)	656.36	1,122.63	487.18	3,158,000 (18.9%)	723.89	1,233.91	545.89	3,390,000 (19.4%)	763.00	1,344.87	572.01	3,592,000 (19.3%)	810.79	1,454.28	619.26
Not in labour force, caring for someone with arthritis	140	19,000 (0.12%)	330.64	363.81	293.53	20,000 (0.12%)	349.92	402.96	303.37	21,000 (0.12%)	362.17	414.24	303.37	22,000 (0.12%)	364.90	429.95	313.34
					Ŵ	Weekly welfare income (AU\$) received by primary carers	e income	(AU\$) rec	eived by p	rimary carei	Ş						
Employed full-time, non-carers	51,455	7,362,000 (46.8%)	25.90	80.88	0.00	8,002,000 (47.8%)	25.78	80.37	0.00	8,549,000 (48.8%)	26.96	82.40	0.00	9,028,000 (48.4%)	26.78	82.83	0.00
Employed part-time, non-carers	22,120	3,116,000 (19.8%)	81.62	142.44	5.75	3,158,000 (18.9%)	82.99	142.81	5.75	3,390,000 (19.4%)	82.96	141.96	5.75	3,592,000 (19.3%)	82.76	142.04	5.75
Not in labour force, caring for someone with arthritis	120	19,000 (0.12%)	240.78	247.98	246.70	18,000 (0.12%)	239.31	243.71	246.70	21,000 (0.12%)	244.93	244.76	246.70	22,000 (0.12%)	243.12	240.89	247.78
					Weekly	Weekly tax paid (includes Medicare levy) (AU\$) by primary carers	ncludes M	edicare le	vy) (AU\$)	by primary (	carers						
Employed full-time, non-carers	51,455	7,362,000 (46.8%)	357.03	496.25	256.09	8,002,000 (47.8%)	378.01	536.25	268.14	8,549,000 (48.8%)	405.27	582.46	282.88	9,028,000 (48.4%)	439.55	628.99	302.73
Employed part-time, non-carers	22,120	3,116,000 (19.8%)	75.81	453.67	3.87	3,158,000 (18.9%)	87.24	501.16	9.59	3,390,000 (19.4%)	92.67	551.66	9.60	3,592,000 (19.3%)	99.59	600.02	14.54
Not in labour force, caring for someone with arthritis	120	19,000 (0.12%)	13.08	87.87	0.00	18,000 (0.12%)	18.02	105.34	0.00	21,000 (0.12%)	18.52	108.88	0.00	22,000 (0.12%)	19.66	114.88	0.00

### Discussion and concluding remarks

## 4.1 Costs of arthritis through lost labour force participation

In Section 2, we estimated that 52,000 people aged 15-64 years were out of the labour force due to arthritis in 2015, and projected that this number would grow to 59,000 in 2030 - a 13% increase. People aged 15-64 years who left the labour force due to arthritis had a significantly lower median income and savings than their full-time employed counterparts with no arthritis in 2015, and the gaps are projected to widen by 2030. The lower total income (and wealth) of those who have lost labour force participation due to arthritis has implications for the economic viability of households. The national impact of arthritis through lost labour force participation among people aged 15-64 years is also projected to increase over the next 15 years, with a 50% increase in lost income, a 23 % increase in welfare payments and a 44% increase in lost taxation revenue. Lost income is expected to grow faster than welfare payments as the indexation of welfare payments is expected to be less than wages growth. Importantly, from a societal cost perspective, there was a 31% increase in lost GDP attributable to arthritis through lost productivity. These are the first projections of the costs of arthritis for individuals, government, and the nation out to 2030.

Arthritis has a significant impact on the working capacity of people with the condition and thus their chances of incurring economic losses<sup>1</sup>. In Australia, the costs of arthritis and other musculoskeletal conditions (including back pain and osteoporosis) were estimated to be \$11.7 billion in 2012, with \$7.4 billion (63.1%) consisting of productivity costs associated with a reduced employment rate (\$6.0 billion) as well as significant costs associated with lost superannuation, presenteeism and absenteeism. Deadweight loss associated with transfer payments (taxation forgone and welfare payments) accounted for a further \$2.3 billion (19.4%) and carer costs were estimated to be 1.2 billion  $(10.4\%)^1$ . The high costs of informal care are indicative of arthritis' deteriorating nature, and the need for individuals with the condition to be assisted and supported; these costs are also expected to increase in the future due to population ageing<sup>20</sup>. Other studies have estimated the indirect costs to be around 80% of the total cost of arthritis<sup>59,60</sup>, and mainly attributable to work disability, absenteeism and lost earnings from employment<sup>61,62</sup>.

However, most of these studies are based on samples that cannot be considered nationally representative, and focus mainly on lost earnings as opposed to considering other relevant economic measures as well<sup>63</sup>. Conversely, the current study uses a sample from a nationally representative population (SDACs 2003, 2009, 2012) and information on income, savings, welfare payments, taxes and GDP for cost projections. The study makes advancements in methodology through developing microsimulation models that integrate outputs from two long-standing microsimulation models (STINMOD, APPSIM) and reliable demographic and labour force projections from Commonwealth Treasury.

In this study, we only estimated the economic costs associated with people who had left the labour force due to their arthritis. This is an under-estimate of the economic costs of lost productivity due to arthritis. Some of those who have arthritis may have moved from full-time employment to part-time employment because of their arthritis. The economic costs of this reduced employment due to arthritis are not captured in the study because of data limitations. While it is possible to identify individuals in full-time or part-time employment with or without arthritis in our main data source (Surveys of Disability, Ageing and Carers), it is not possible to identify if they are in part-time employment because of their arthritis or some other reason.

Investment in preventive health is recognised as essential to overcoming the detrimental impacts of ill-health on labour force participation<sup>64</sup>. A number of randomised controlled trials have demonstrated that arthritis treatments can increase labour force participation<sup>65</sup>. Studies have also demonstrated that workplace changes can reduce work disability from arthritis<sup>66</sup> and thus help to retain employment of those with arthritis. A supportive work environment such as those permitting adjusted working hours, adjusted job demands, accessibility of workplace and supportive co-workers and employers can reduce withdrawal from the labour force by people with arthritis<sup>67</sup>. Lacaille et al. (2004) also found that modifying work-related factors that increase the risk of work disability in people with arthritis can increase the participation of sufferers<sup>68</sup>. In Australia, self-management interventions to overcome workplace challenges associated with chronic physical pain (such as arthritis) have been shown to be effective. An example is the 'ADAPT' program for work-related pain, which is

### Discussion and concluding remarks

an intensive cognitive-behavioural pain management program run by the Pain Management and Research Centre (University of Sydney) and Royal North Shore Hospital (http://sydney.edu.au/medicine/pmri/patientservices/resources/index.php). For those who are likely to exit from the labour force due to their arthritis, providing counselling and training, and assisting them to find new jobs which are less physical in nature, where possible, can help them to stay in the labour market<sup>69</sup>. These types of interventions are likely not only to help increase labour force participation and economic growth<sup>39,70</sup> but ensure future governments have sufficient revenue for meeting the healthcare needs of the ageing population<sup>39</sup>.

### 4.2 Costs of arthritis through informal caregiving

We estimated that 41,000 people aged 15-64 years were caring for someone with arthritis and, of these, 19,000 had left the labour force due to their caring responsibilities in 2015. This number is projected to increase to 22,000 in 2030 – a 16% increase. People aged 15-64 years who are not in the labour force and providing care for someone with arthritis had a lower median income than non-carers working full-time or part-time, with income differences widening over time. We are currently developing a microsimulation model to estimate the national costs of lost productivity from caregiving.

Previous governments have focused on increasing labour force participation rates using economic incentives that target particular subgroups of the working-age population. For example, new parents can access the Commonwealth Government's paid parental leave which provides financial support for eligible parents for up to 18 weeks after the birth or adoption of their child. For older workers, the 15% tax on lump sums and pensions from superannuation schemes after the age of 60 years has been removed<sup>71</sup>, which encourages continuation in work. There is also the Age Discrimination Act 2004 which provides job protection for all workers in Australia<sup>72</sup>. However, these "broad brush" approaches to help people either return to, or remain in, paid work fail to take into account one of the main reasons people often have to leave their jobs quickly - to take on the caring needs of a relative or friend. Moreover, more

needs to be achieved in terms of policy design. There needs to be greater consideration given to the main chronic/disabling conditions associated with most of the lost labour force participation among carers (e.g. back pain, arthritis). Until then, these incentives will not have a major impact on the labour force participation of informal carers.

Conventionally, Australian health policy has focused on the delivery of healthcare to improve the health of citizens for its own sake and employment policy has mainly been conducted in isolation from health policy. Recent health reforms, however, seem to encapsulate the view that "good health policy is part of good economic policy" as suggested in Russell et al. (2008)<sup>64</sup>. This philosophy naturally leads to the necessity of addressing Australia's increasing burden of chronic conditions with the highest care demands.

Overall, the main challenges faced by carers seem to be due to pressures already in the health system and the lack of effective measures for ensuring workplace flexibility for carers. Whilst there are public and private care services for those with a disability, chronic condition, or frail aged (such as residential and aged care facilities) and respite care for carers, available services are insufficient to meet demand, resulting in delayed or constrained access<sup>73</sup>. Moreover, the policies adopted in the last 20 years have seen a substantial move away from institutional forms of care to 'ageing in place' i.e. community-based care (http://ageinplace.com/ aging-in-place-basics/what-is-aging-in-place/) which can place greater pressure on informal carers.

Employment policies in relation to working carers having suitable forms of support in their workplaces may also need further attention. Working carers are protected from discrimination when attempting to balance work schedules with family and caring responsibilities. Under the Equal Opportunity Act 2010, employers have a positive responsibility to undertake practical and comparable measures to remove discrimination, sexual harassment and victimisation from their workplace as much as possible. The Act applies to employers (organisations) of all sizes, includes all types of workers, and applies to all stages of employment. Although there are legal protections in place for carers to be in a position to manage their work and caring responsibilities effectively<sup>74</sup>, challenges remain in relation to whether (a) workers feel they are able to discuss any difficulties

## Discussion and concluding remarks

they face with employers when they occur, and (b) employers are feasibly able to provide the type of workplace flexibility (flexible work hours, part-time work or paid carer leave) required to meet the needs of their workers who are also carers.

As demonstrated in this report on arthritis, informal caregiving has negative effects on carers, such as reduced labour force participation and lower income. Previous studies have shown that male and female carers also differ in the way they strive to manage their work and caring duties. Women are more likely to reduce their hours of work or stop working altogether when becoming a carer<sup>75</sup> and thus female carers are more likely to incur additional risks (and costs) in terms of maintaining a good career trajectory and income, compared to male carers<sup>56,58</sup>.

With persistent skills shortages and the ageing population needing more healthcare and support in the future, the government will need to adopt a more holistic approach when endeavouring to increase labour force participation in Australia. Addressing some of the challenges effecting working carers and carers wanting to work (http://www.workingcarers.org.au/) may help to improve labour force participation among this growing group.

### Labour force gains and the associated

economic gains if people with osteoarthritis of the knee could be engaged in the ADAPT trial strategies

The Arthritis, Diet, and Activity Promotion Trial (ADAPT) is one of the best studies available to demonstrate the benefits of interventions including both diet and exercise management for knee osteoarthritis. Consequently this section focuses on people with osteoarthritis of the knee only although they make up a relatively small part of the total arthritis population. We simulated the potential labour force gains and the associated economic gains of managing knee OA, as a case study, to demonstrate that interventions to manage knee OA (or other chronic conditions) have the potential to provide economic benefits both to individuals and the government in addition to health benefits.

Osteoarthritis (OA) of the knee is a chronic condition that is strongly related to obesity<sup>76</sup>, is most common among older women<sup>77</sup>, and a major source of physical disability with ageing<sup>78,79</sup>. The main symptoms of knee OA are pain in the joints and stiffness<sup>79</sup>. Because there is no known remedy for this condition, much of the focus in public health and medicine is on palliative care i.e. the aim of the treatment is to improve or to preserve quality of life related to health. In recent years, there has been cumulative evidence that physical activity is an effective treatment for the reduction of pain and improves the physical function of older people with knee OA; for example a recent Cochrane review of 32 RCTs involving nearly 3,800 men and women found improvements in self-reported knee pain and function of patients who participated in on-the-ground exercises<sup>81</sup>. The Arthritis, Diet, and Activity Promotion Trial (ADAPT) is one such study that examined the combined impact of weight loss and physical activity on the functioning of overweight and obese older adults with knee OA4,24. Adding to this work, the study by Rejeski et al (2002)<sup>5</sup> demonstrates the impact of the interventions in ADAPT on quality of life for older obese people with knee OA. Specifically, Rejeski et al (2002)<sup>5</sup> reported on how dietary weight loss and exercise influence the combined mental and physical health scores of the SF-36 as well as measures of satisfaction with physical function and appearance. (The SF-36 is a generic measure of guality of life related to health, whereas the satisfaction measures assess specific end points that relate theoretically to the intervention and the population under examination.) Secondary analyses were conducted on the eight subscales of the SF-36.

In the Arthritis, Diet, and Activity Promotion Trial  $(ADAPT)^{4,5}$ , a total of 316 older men (28.03%) and women (71.97%) (with mean age = 68.52 years (SD 6.30)) were randomised to one of four 18-month long treatments:

- exercise only
- dietary weight loss only
- dietary weight loss plus exercise
- a healthy lifestyle control condition

It was found that the combined dietary weight loss plus exercise intervention had the most consistent, positive effect on quality of life outcomes compared with the control group; however, the findings were limited to measures of physical health or psychological outcomes related to the physical self (see Table 3 in Rejeski et al (2002)).

In this section, we project the potential labour force benefits of managing knee OA by implementing the ADAPT intervention<sup>5</sup>. Pain reduction among people with knee OA with the use of the three strategies noted above (compared to the control) is reported via the SF-36. The projected benefits are derived for the working-age population (aged 15-64 years) and reflect the probability of being in the labour force associated with severity of pain experienced. Furthermore, as it is known that premature or unplanned exits from the workforce are associated with significant costs to both individuals and governments<sup>19</sup>, we also project the follow-on benefits associated with the potentially increased labour force participation to individuals in terms of increased income, and to the government in terms of increased income tax revenue and reduced welfare payments.

Because there is no direct measure of the effect of the intervention on labour force participation in the ADAPT series<sup>5</sup>, we model improvement in the labour force participation rates by analysing the effects of managing arthritis by implementing the different arms of the trial on individuals' quality of life (i.e. the SF-36 bodily pain domain score)<sup>28</sup>. The Household, Income and Labour Dynamics in Australia (HILDA) Survey data<sup>29</sup> are then used to model the association between SF-36 bodily pain scores and labour force participation.

### Labour force gains and the associated

economic gains if people with osteoarthritis of the knee could be engaged in the ADAPT trial strategies

All of this information is then incorporated into our microsimulation models (Health&WealthMOD and Helath&WealthMOD2030) to estimate the potential increase in the labour force associated with the improvement in SF-36 scores of the individuals with knee OA if the ADAPT program is implemented. Using these data, together with the estimated number of Australians aged 15-64 years who left the labour force due to their knee OA, we simulated how many of them would have continued to stay in the labour force if their knee OA was managed using the ADAPT interventions.

#### 5.1 Data and methods

Published results from the ADAPT study (particularly Rejeski et al (2002)<sup>5</sup> and nationally representative Australian household surveys were used to estimate the potential labour force and related economic benefits associated with pain reduction in patients with knee OA via participation in ADAPT. Although the trial collected some information on the socio-economic characteristics of participants at baseline (income, education), labour force participation was not among the characteristics included, as the trial was not designed to analyse the effect of the different management strategies for knee OA on labour force participation.

Since it was not possible to directly estimate the effect of the different management strategies on labour force participation using the RCT data, the effects on labour force participation were estimated based on the association between improvements in guality of life, as a result of taking up one of the strategies, and labour force participation. The SF-36 bodily pain domain score was used as the quality of life measure. The SF-36 bodily pain domain is one of the eight domains of the Medical Outcomes Study Short Form-36 Health Survey (SF-36), with other domains being physical functioning, role limitations-physical, social functioning, general health, role limitations-emotional, mental health and vitality<sup>28</sup>. Each domain score ranges from 0 to 100, with higher scores reflecting better health status. The SF-36 bodily pain domain is used to evaluate overall pain.

#### 5.1.1 Effectiveness of interventions

The effectiveness estimates of the three management strategies (dietary weight loss, exercise, dietary weight loss plus exercise) on improving pain were based on the SF-36 bodily pain domain results of the ADAPT study published by Rejeski et al (2002)<sup>5</sup>. Compared to the control group, improvements in the SF-36 bodily pain score from baseline to follow-up (adjusted for covariates including the pre-randomisation levels of the outcome variable, age, and gender) were significantly greater at the p<.01 level for the dietary weight loss plus exercise intervention only. The improvement in SF-36 bodily pain score from baseline to follow up for the dietary weight loss plus exercise intervention splus exercise group was 4.73 units (i.e. the effect size) greater than for the control group.

### 5.1.2 Association between SF-36 bodily pain scores and labour force participation

The association between the SF-36 bodily pain score and the probability of being in the labour force were analysed using Wave 10 data of The Household Income and Labour Dynamics in Australia (HILDA) Survey which is a nationally representative Australian household panel survey<sup>29</sup>. Analysis was restricted to the surveyed populations aged 20-64 years and those who had SF-36 bodily pain scores in the range between 0 and 70. This would exclude individuals in the HILDA Survey who had lesser or no pain (i.e. higher SF-36 bodily pain score) and reduce the chance of contaminating the association between the SF-36 bodily pain scores and the probability of being in the labour force.

The association between the SF-36 bodily pain score and labour force participation was estimated as the prevalence ratio (ratio of the probability of being in the labour force) associated with each unit increase in the SF-36 bodily pain score using a modified Poisson regression model with a robust error variance to estimate the prevalence ratios<sup>82</sup>. Analyses were adjusted for the confounding effects of age and highest level of education, and the models were separately fitted for men and women.

### Labour force gains and the associated

economic gains if people with osteoarthritis of the knee could be engaged in the ADAPT trial strategies

#### 5.1.3 Labour force impacts

We estimated the number of persons, who could have continued to stay in the labour force, among those not in the labour force due to their knee OA, as a result of pain reduction if one of the interventions (dietary weight loss only, exercise only, or dietary weight loss plus exercise) were in place for those with knee OA. This was estimated as the potential increase in those who were in the labour force and have knee OA using the equation:

$$AddLF = Num_{int} * LF_{kneeQA} * LF_{Growth}$$
(1)

where  $Num_{int}$  is the number of people with knee OA participating in the ADAPT intervention<sup>5</sup>;  $LF_{kneeOA}$  is the estimated improvement in labour force participation of those with knee OA (a rate); and  $LF_{Growth}$  is the estimated growth in the labour force participation rates associated with pain reduction as a result of participating in one of the interventions for individuals with knee OA. This is estimated as

$$LF_{Crowth} = \exp(^{Effect^*\Delta Pain}) - 1$$
 (2)

where *Effect* is the estimated association between the labour force participation and SF-36 bodily pain domain score for each unit change in SF-36 bodily pain domain score, such that  $exp^{Effect}$  is a prevalence ratio of being in the labour force associated with each unit change in SF-36 bodily pain domain score; and  $\Delta Pain$  is the estimated improvement in SF-36 bodily pain score for the patients undergoing one of the interventions over the control group.

#### 5.1.4 Economic benefits

The economic benefits of the potential increase in the number of people in the labour force who have knee OA as a result of pain reduction through uptake of one of the interventions in ADAPT were estimated as discussed in Stage 1 (See Section 2.4). The economic benefits were estimated at the national level. For individuals, it was estimated in terms of an increase in their total accumulated income for all additional people in the labour force, and for the government, in terms of savings in welfare (transfer) payments and additional income tax revenue. This modelling is limited to people already out of the workforce due to their arthritis. It is also likely that the ADAPT interventions, if offered to employed people at risk of leaving the labour force due to their knee osteoarthritis, would help to avoid or delay their exit from the labour force. However, due to data limitations, this scenario could not be modelled. Consequently, estimates of the economic benefits of this intervention are likely to underestimate the benefits.

#### 5.2 Results

#### 5.2.1 Not in the labour force due to knee OA

The prevalence data and the labour force information were not available for those with OA and knee specific OA in our main data sources, the Surveys of Disability, Ageing and Carers, which only have information for those with overall arthritis. Thus, we estimated the labour force information for those with knee specific OA using the projected proportion of OA among overall arthritis and the results from a study which analysed the data on OA related GP presentations from the Bettering the Evaluation and Care of Health (BEACH) program<sup>83</sup>. The study reported the number of GP presentations for knee specific OA and OA overall by age groups allowing us to estimate the age-specific proportion of knee specific OA among those with OA overall. We then further estimated the proportion of knee specific OA among those for arthritis overall using the projected prevalence data on both OA and arthritis overall. Assuming all individual labour force groups would have the same proportion of knee specific OA among those with arthritis, the estimated number of those who were not in the labour force due to their knee OA would be 7,605 in 2015 rising to 8,743 by 2030 (Table 5).

### Labour force gains and the associated

economic gains if people with osteoarthritis of the knee could be engaged in the ADAPT trial strategies

Table 5: Projected numbers of those not in the labour force due to knee OA, aged 15-64 years

Gender	2015	2020	2025	2030
Male	2,583	2,807	2,910	2,965
Female	5,022	5,370	5,529	5,778
Total	7,605	8,177	8,439	8,743

### 5.2.2 Association between SF-36 bodily pain scores and labour force participation

For those aged between 20 to 64 years old and with bodily pain domain scores between 0 and 70, the estimated prevalence ratios of being in the labour force associated with each unit increase in SF-36 bodily pain domain score was 1.0112 for men and 1.0103 for women (Table 6). Thus for a 11.86 unit estimated improvement in SF-36 bodily pain score due to *dietary weight loss plus exercise* intervention from baseline to follow-up, there would be an estimated increase of 14% in the labour force participation rate of men and 12.9% for women with knee OA by follow-up (Table 6).

We also estimated the increase in the labour force participation rate due to the dietary weight loss plus exercise intervention versus *healthy lifestyle only* (the control group in the ADAPT Study). There was an estimated improvement of 4.73 units in SF-36 bodily pain domain score from baseline to follow up for the dietary weight loss plus exercise intervention group over the healthy lifestyle control group. Thus, managing of knee OA with dietary weight loss plus exercise intervention would increase the labour force participation rate by approximately 5.4% in men and 4.9% in women compared to managing it with a healthy lifestyle (Table 6). Table 6: Percentage increase in the labour force participation rates associated with pain reduction due to Diet plus Exercise intervention for the management of knee OA in the ADAPT study

	Male	Female
Prevalence ratios of being in the labour force associated with each unit increase in SF-36 bodily pain score	1.0112	1.0103
Estimated percentage increase in the labour force participation rate from baseline to follow-up with dietary weight loss plus exercise intervention	14%	12.9%
Estimated percentage increase in the labour force participation rate from baseline to follow- up with dietary weight loss plus exercise intervention over healthy lifestyle intervention (control group)	5.4%	4.9%

### Labour force gains and the associated

economic gains if people with osteoarthritis of the knee could be engaged in the ADAPT trial strategies

#### 5.2.3 Impacts on labour force participation

We simulated the labour force gains and the associated economic benefits of managing knee OA with a dietary weight loss plus exercise intervention compared to not managing at all and compared to managing it with a healthy lifestyle intervention (control group in the ADAPT study).

If those not in the labour force due to their knee OA (Table 5) had taken up the dietary weight loss plus exercise intervention to manage their knee OA, it is estimated that there would be an additional 231 men and 341 women aged 15-64 years in the labour force,

who otherwise would be out of the labour force due to their knee OA in 2015 (Table 7). By 2030, the number of men and women who would have avoided being out of the labour force due to their knee OA would have increased to 284 for men and 431 for women.

If knee OA was managed using dietary weight loss plus exercise intervention compared to managing it with a healthy lifestyle intervention, the estimated number of people who would have avoided being out of the labour force due to their knee OA as a result of reduced pain would be 88 for men and 131 for women in 2015 rising to 109 for men and 166 for women in 2030.

Table 7: Estimated number of people who would have avoided being out of the labour force due to their knee OA as a result of dietary weight loss plus exercise intervention for the management of their knee OA, 15-64 years old

Gender	2015	2020	2025	2030
Dietary weight lo	oss plus exercise interv	ention		
Male	231	259	277	284
Female	341	383	407	431
Dietary weight lo	oss plus exercise interv	ention versus healthy	life style (control grou	(qı
Male	88	99	106	109
Female	131	147	157	166

#### 5.2.4 Economic impacts

The potential increase in the labour force of those with knee OA as a result of reduced pain by managing their knee OA using ADAPT interventions has follow-on economic benefits both to individuals in terms of an increase in their incomes and for the government in terms of savings in welfare payments and additional income tax revenue.

Adopting a dietary weight loss plus exercise intervention as described in the ADAPT trial would result in a total increase of \$20.8 million in personal income per year associated with people avoiding being out of the labour force due to their knee OA. (Table 8). Similarly, it is estimated that there would be an associated saving of about \$7.6 million per year in welfare payments and an increase of \$5.4 million in income tax revenue. By 2030, the economic benefits would have increased to an extra \$33.5 million in income per year, a reduction of \$9.9 million per year in welfare payments, and an increase of \$8.4 million in income tax revenue (Table 8).

### Labour force gains and the associated

economic gains if people with osteoarthritis of the knee could be engaged in the ADAPT trial strategies

Table 8: Estimated economic benefits from increased labour force associated with reduced pain as a result of managing their knee OA with dietary weight loss plus exercise intervention, 15-64 years old (in thousands, in 2015 AU\$)

	2015	2020	2025	2030
Dietary weight loss plus exercise inte	rvention			
Increase in annual incomes	20,770	25,088	29,624	33,488
Increase in annual income tax payment	5,430	6,469	7,515	8,399
Reduction in annual welfare payment	7,578	8,401	9,240	9,909
Dietary weight loss plus exercise inte	rvention versus h	ealthy life style (	control group)	
Increase in annual incomes	7,918	9,594	11,392	12,859
Increase in annual income tax payment	2,070	2,473	2,890	3,218
Reduction in annual welfare payment	2,896	3,224	3,549	3,812

If knee OA is managed using a dietary weight loss plus exercise intervention compared to managing it with a healthy lifestyle intervention as described in the ADAPT study, it is estimated that the number of people who would have avoided being out of the labour force due to their knee OA would be 88 for men and 131 for women in 2015 (Table 7). As a result of this increase in the labour force, there would be a total estimated increase of \$7.9 million in income per year, an estimated increase of \$2.1 million in income tax revenue and a reduction of \$2.9 million in welfare payments per year (Table 8). By 2030 this would have increased to a total estimated increase of \$12.9 million in income per year, an estimated increase of \$3.2 million in income tax revenue and a reduction of \$3.8 million in welfare payments per year.

# Conclusions

#### Stage 1

In Section 3, we projected that the economic costs of lost productivity due to arthritis in Australians age 15-64 years would increase by 50% in lost income, 23% in additional welfare payments, and 44% in lost taxation revenue from 2015 to 2030. Moreover, these lost workers due to arthritis resulted in a loss of \$7.2 billion in GDP in 2015, increasing to \$9.4 billion in 2030.

In Section 4, we projected the number of people aged 15-64 years who are out of the labour force due to caring for someone with arthritis to increase from 19,000 in 2015 to 22,000 in 2030. At both time points, carers received less income, more welfare payments, and paid less tax than people employed full-time (or part-time) and who were non-carers. Given the economic challenges facing Australia and other developed countries (such as severe skills shortages, rising healthcare costs, the ageing population), successive governments will need to adopt holistic approaches to reducing the costs of informal care through lost labour force participation.

#### Stage 2

People with knee OA who experience less pain are more likely to be in the labour force. Effective interventions that reduce bodily pain may thus help patients to remain in the workforce longer. It has been estimated that, among people who had left the labour force due to their knee OA, 572 people aged 15-64 years could have remained in the labour force in 2015 if they had participated in a dietary weight loss plus exercise intervention, based on the ADAPT trial data. The cumulative economic benefit of this increased labour force participation would be an estimated increase of \$20.8 million in their annual income in 2015, which would increase to \$33.5 million per year in 2030. The benefit of increased labour force participation associated with pain reduction by managing knee OA through the ADAPT intervention extends beyond individuals, and provides economic benefits to government too. We estimated a decrease in the number of people with knee OA receiving welfare payments and an increase in the number of people paying income tax.



- 1. Arthritis and Osteoporosis Victoria. A problem worth solving. Elsternwick: Arthritis and Osteoporosis Victoria, 2013.
- 2. Australian Institute of Health and Welfare. Health-care expenditure on arthritis and other musculoskeletal conditions 2008-09. Arthritis series 20, Cat no PHE 177. Canberra: AIHW, 2014.
- 3. Dunlop D, Manheim L, Yelin E, et al. The costs of arthritis. Arthritis & Rheumatism 2003;49(1):101-13.
- 4. Messier SP, Loeser RF, Miller GD, et al. Exercise and Dietary Weight Loss in Overweight and Obese Older Adults With Knee Osteoarthritis: The Arthritis, Diet, and Activity Promotion Trial. Arthritis & Rheumatism 2004;**50**(5):1501–10.
- 5. Rejeski WJ, Focht BC, Messier SP, et al. Obese, Older Adults With Knee Osteoarthritis: Weight Loss, Exercise, and Quality of Life. Health Psychology 2002;**21**(5):419–26.
- Global Burden of Disease Study 2013 Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. The Lancet 2015 22 August;**386**(9995):743-800.
- 7. World Health Organisation. The impact of chronic disease in high income countries. Preventing Chronic Disease: A vital investment. Geneva: World Health Organisation, 2005.
- 8. World Health Organisation. Preventing chronic diseases: a vital investment. Geneva: World Health Organisation, 2005.
- 9. Organisation for Economic Co-operation and Development. Sickness, disability and work: breaking the barriers. Paris: OECD Publishing, 2007.
- 10. Commonwealth of Australia. 2015 Intergenerational Report: Australia in 2055. Canberra: The Commonwealth of Australia, 2015.
- 11. Centers for Disease Control and Prevention. Addressing the Nation's Most Common Cause of Disability At A Glance 2015 Atlanta: National Center for Chronic Disease Prevention and Health Promotion; 2015 [Available from: http://www.cdc.gov/chronicdisease/resources/publications/aag/arthritis.htm].
- 12. Wong R, Davis AM, Badley E, et al. Prevalence of Arthritis and Rheumatic Diseases around the World: A Growing Burden and Implications for Health Care Needs. Toronto: Arthritis Community Research and Evaluation Unit, 2010.
- 13. Australian Institute of Health and Welfare. Arthritis, osteoporosis and other musculoskeletal conditions 2015 [Available from: http://www.aihw.gov.au/arthritis-and-musculoskeletal-conditions/].
- 14. Australian Institute of Health and Welfare. Australia's Health, 2010. Canberra: AIHW, 2010.
- Center for Disease Control. National and State Medical Expenditures and Lost Earnings Attributable to Arthritis and Other Rheumatic Conditions - United States, 2003. Morbidity and Mortality Weekly Report (MMWR) 2007;56(01):4-7.
- 16. Australian Institute of Health and Welfare. Australia's Health, 2008. Canberra: AIHW, 2008.
- 17. Hunter DJ, Schofield D, Callander E. The individual and socioeconomic impact of osteoarthritis. Nature Reviews Rheumatology 2014;**10**:437–41.

- 18. Gignac MA, Cao X, Lacaille D, et al. Arthritis-Related Work Transitions: A Prospective Analysis of Reported Productivity Losses, Work Changes, and Leaving the Labor Force. Arthritis & Rheumatism 2008;**59**(12):1805-13.
- 19. Schofield D, Shrestha R, Percival R, et al. The personal and national costs of lost labour force participation due to arthritis: an economic study BMC Public Health 2013;13:1-10.
- 20. Schofield D, Shrestha R, Cunich M, et al. Lost productive life years caused by chronic conditions in Australians aged 45-64 years, 2010 to 2030. Medical Journal of Australia 2015;203(6):260.e1-6.
- 21. Li X, Gignac MA, Anis AH. The indirect costs of arthritis resulting from unemployment, reduced performance, and occupational changes while at work. Medical Care 2006;**April 44**(4):304-10.
- 22. Schofield D, Cunich M, Shrestha R, et al. The impact of chronic conditions of care recipients on the labour force participation of informal carers in Australia: which conditions are associated with higher rates of non-participation in the labour force? BMC Public Health 2014;14(1):1-9.
- 23. Schofield DJ, Callander EJ, Shrestha RN, et al. Labour force participation and the influence of having arthritis on financial status. Rheumatology International July 2015;35(7):1175-81.
- 24. Sevick MA, Miller GD, Loeser RF, et al. Cost-Effectiveness of Exercise and Diet in Overweight and Obese Adults with Knee Osteoarthritis. Medicine & Science in Sports & Exercise 2009;**41**(6):1167-74.
- 25. Arthritis Australia. Time to Move: Arthritis. Sydney: Arthritis Australia, March 2014.
- Schofield D, Shrestha R, Callander E, et al. Modelling the cost of ill health in Health&WealthMOD (Version II): lost labour force participation, income and taxation, the impact of disease prevention. The International Journal of Microsimulation 2011;4(3):32-36.
- Schofield D, Shrestha R, Kelly S, et al. A Microsimulation Model of the Long Term Economic Impacts of Disease on the Labour Force Participation of Australians Aged 45-64 years: What are the Costs to Individuals and Government and the Opportunities for Effective Interventions? International Journal of Microsimulation 2014;7(2):94-118.
- Ware JE, Jr., Sherbourne CD. The MOS 36-item short-form health survey (SF-36). I. Conceptual framework and item selection. Med Care 1992;30(6):473-83.
- 29. Summerfield M, Freidin S, Hahn M, et al. HILDA User Manual Release 11: Melbourne Institue of Applied Economic and Social Research, University of Melbourne, 2012.
- 30. Australian Bureau of Statistics. Information Paper Basic Confidentialised Unit Record File: Survey of Disability, Ageing and Carers 2003 (reissue). Canberra: Australian Bureau of Statistics, 2005.
- 31. Australian Bureau of Statistics. Information Paper: Disability, Ageing and Carers, Australia: User Guide 2009, Cat 4431.0.55.001. Canberra: Australian Bureau of Statistics, 2011.
- 32. Australian Bureau of Statistics. Disability, Ageing and Carers, Australia: Summary of Findings, 2012, 2013.
- 33. Percival R, Abello A, Vu QN. STINMOD (Static Income Model) 2007. In: Gupta A, Harding A, eds. Modelling Our Future: Population ageing, health and aged care. Amsterdam: Elsevier B.V., 2007.
- 34. Lambert S, Percival R, Schofield D, et al. An Introduction to STINMOD: A Static Microsimulation Model. Canberra: NATSEM, 1994.
- 35. Percival R. Building STINMOD's base population Canberra: NATSEM, 1994.
- 36. Keegan M, Kelly S. APPSIM Dynamic microsimulation modelling of social security and taxation. Canberra: National Centre for Social and Economic Modelling, University of Canberra, 2009.

- 37. Bell P. GREGWT and TABLE macros users guide. Canberra: Australian Bureau of Statistics, 2000.
- 38. Treasury. Budget Paper No 1: Budget Strategy and Outlook 2015-16. Canberra: Commonwealth Treasury, 2015.
- 39. Commonwealth Treasury. Intergenerational report 2007. Canberra: Commonwealth of Australia, 2007.
- 40. Percival R, Kelly S. Who's going to care? Informal care and an ageing population Canberra: NATSEM, 2004.
- 41. Disability Rights Commission, Equal Opportunities Commission, Carers UK. The Future: Who Carers?: University of Leeds, 2006.
- 42. Pickard L, Wittenberg R, Comas-Herrara A, et al. Care by Spouses, Care by Children: Projections of Informal Care for Older People in England to 2031. Social Policy and Society 2007;6(3):353-66.
- 43. Kelland K. Dementia to drive global leap in number of elderly needing care. London: Reuters, 2013.
- 44. Productivity Commission. Caring for Older Australians Report No 53, Final Inquiry Report. Canberra: Productivity Commission, 2011.
- 45. Carmichael F, Charles S. The opportunity costs of informal care: does gender matter? Journal of Health Economics 2003;22(5):781-803.
- 46. Australian Bureau of Statistics. Survey of Disability, Ageing and Carers 2009 Canberra: Australian Bureau of Statistics, 2012.
- 47. Begg SJ, Vos T, Barker B, et al. Burden of disease and injury in Australia in the new millennium: measuring health loss from diseases, injuries and risk factors. Med J Aust 2008;**188**(1):36-40.
- 48. Schofield D, Passey M, Earnest A, et al. Are We Getting Healthier As We Grow Older? Implications for Babyboomer Labor Force Participation. Ann NY Acad Sci 2007;**1114**:230-40.
- 49. Lancaster P, Huang J, Pedisich E. Australia's mothers and babies 1991. ANPS Unit. Canberra: Australian Institute of Health and Welfare (AIHW), 1994.
- 50. Li Z, McNally L, Hilder L, et al. Australia's mothers and babies 2009 Perinatal statistics series no 25 Cat no PER 52. Sydney: AIHW National Perinatal Epidemiology and Statistics Unit, 2011.
- 51. Doyle L, Roberts G, PJ A. Outcomes at Age 2 Years of Infants < 28 Weeks' Gestational Age Born in Victoria in 2005. Pediatrics 2010;**156**(1):49-53.
- 52. Pavalko EK, Artis JE. Women's Caregiving and Paid Work: Causal relationships in late midlife. The Journals of Gerontology 1997;**52b**(4s):170-9.
- 53. Speiss CK, Schnieder U. Interactions between care-giving and paid work hours. Ageing and Society 2003; 23:41-68.
- 54. Heitmueller A, Michaud P. Informal Care and Employment in England: Evidence from the British Household Panel Survey, IZA Discussion Paper 2010. Bonn: IZA, 2006.
- 55. Lee C, Gramotnev H. Transitions into and out of caregiving: Health and social characteristics of mid-age Australian women. Psychology & Health 2007;**22**(2):193-209.
- 56. Schofield D, Shrestha R, Callander E, et al. Costs of being a carer: labour force participation and lost earnings among older working-aged Australians Australian and New Zealand Journal of Public Health [Letters] 2013;**37**(2):192-3.
- 57. Nguyen HT, Connelly LB. The effect of unpaid caregiving intensity on labour force participation: results from a multinomial endogenous treatment mode. Social Science & Medicine 2014;100:115-22.

- 58. Bittman M, Hill T, Thomson C. The Impact of Caring on Informal Carers' Employment, Income and Earnings: A Longitudinal Approach. Australian Journal of Social Issues 2007;2(2):255-72.
- 59. Chen A, Gupte C, Akhtar K, et al. The Global Economic Cost of Osteoarthritis: How the UK Compares. Arthrtis 2012;2012.
- 60. Rat A, Boissier M. Rheumatoid arthritis: direct and indirect costs. Joint Bone Spine 2004;71(6):518-24.
- 61. Wolfe F, Michaud K, Choi HK, et al. Household income and earnings losses among 6,396 persons with rheumatoid arthritis. Journal of Rheumatology 2005;**32**(10):1875-83.
- 62. Puolakka K, Kautiainen H, Pekurinen M, et al. Monetary value of lost productivity over a five year follow up in early rheumatoid arthritis estimated on the basis of official register data on patients' sickness absence and gross income: experience from the FIN-RACo trial. Ann Rheum Dis 2006 Jul;65(7):899-904. Epub 2005 Nov 16.
- 63. Allaire S, Wolfe F, Niu J, et al. Work disability and its economic effect on 55-64-year-old adults with rheumatoid arthritis Arthritis Care & Research 2005;53(4):603-08.
- 64. Russell L, Rubin G, Leeder S. Preventive health reform: what does it mean for public health? MJA [Debate] 2008;188(12):4.
- Yelin EH, Trupin LS, Sebesta DS. Transitions in employment, morbidity, and disability among persons ages 51–61 with musculoskeletal and non-musculoskeletal conditions in the US, 1992–1994. Arthritis Rheum 1999;42 (769-779).
- 66. Shanahan EM, Ahern M. Inflammatory arthritis and work disability: what is the role of occupational medicine? Occup Med (Lond) 2008;**58**(1):2-4.
- 67. Chorus AM, Miedema HS, Wevers CW, et al. Work factors and behavioural coping in relation to withdrawal from the labour force in patients with rheumatoid arthritis. Ann Rheum Dis 2001;60(11):1025-32.
- 68. Lacaille D, Sheps S, Spinelli JJ, et al. Identification of Modifiable Work-Related Factors That Influence the Risk of Work Disability in Rheumatoid Arthritis. Arthritis & Rheumatism 2004;**51**(5):843-52.
- 69. Arthritis Australia. Submission on the Government's consultation paper on Disability and Employment. Sydney: Arthritis Australia, 2013.
- 70. Council of Australian Governments. Council of Australian Governments' meeting 10 February 2006 communique 2006 [Available from: www.treasury.gov.au/documents/1221/PDF/02\_NRA.pdf
- 71. Australian Government. A Plan to Simplify and Streamline Superannuation (Accessed 20 July 2006), http://www.budget.gov.au/, 2006.
- 72. Australian Human Rights Commission. Legislation: Age Discrimination Act 2004 2013 [Available from: http://www.humanrights.gov.au/our-work/legal/legislation.
- 73. Holland KE. Carers' perspectives on caring: A qualitative analysis of open-ended responses to the Carer Health and Wellbeing Index survey 2007. Canberra: Carers Australia, 2008.
- 74. Victorian Equal Opportunity and Human Rights Commission. Employee rights with carer status, family responsibilities and parental status Melbourne: Victorian Equal Opportunity and Human Rights Commission; 2015.
- 75. Council of Australian Governments (COAG) Reform Council. Chapter 6: Caring responsibilities. Tracking equity: Comparing outcomes for women and girls across Australia. Canberra: COAG Reform Council, 2013.

- 76. Bliddal H, Leeds AR, Christensen R. Osteoarthritis, obesity and weight loss: evidence, hypotheses and horizons a scoping review. Obes Rev 2014;15(7):578–86.
- 77. Australian institute of health and welfare. A snapshot of arthritis in Australia 2010. Canberra: AIHW, 2010.
- 78. Australian institute of health and welfare. A picture of osteoarthritis in Australia Canberra: AIHW, 2007.
- 79. Felson DT, Naimark A, Anderson J, et al. The prevalence of knee osteoarthritis in the elderly. the framingham osteoarthritis study. Arthritis & Rheumatism 1987;**30**(8).
- 80. Martin SN, McGowan HJ, Smith RF. Exercise for the Treatment of Knee Osteoarthritis. Am Fam Physician 2011;84(5):1.
- 81. Fransen M, McConnell S. Exercise for osteoarthritis of the knee. Cochrane Database Syst Rev 2008;4(CD004376).
- 82. Zou G. A modified poisson regression approach to prospective studies with binary data. Am J Epidemiol 2004;159(7):702-6.
- 83. Brand CA, Harrison C, Tropea J, et al. Management of osteoarthritis in general practice in Australia. Arthritis Care Res (Hoboken) 2014;66(4):551-8.



PO Box 550 Broadway NSW 2007 www.arthritisaustralia.com.au